Safety Science 93 (2017) 29-49

Contents lists available at ScienceDirect

Safety Science

journal homepage: www.elsevier.com/locate/ssci

Agent-based modelling and mental simulation for resilience engineering in air transport



Netherlands Aerospace Centre NLR, Safety Institute, Anthony Fokkerweg 2, 1059 CM Amsterdam, The Netherlands

ARTICLE INFO

Article history: Received 4 August 2016 Received in revised form 1 November 2016 Accepted 3 November 2016

Keywords: Agent-based modelling and simulation Resilience Air traffic Qualitative model Mental simulation

ABSTRACT

Following a discussion of recent reviews, we argue that in resilience engineering (RE) there is a need for more structured modelling approaches for analysis of resilience in sociotechnical systems that can support both qualitative and quantitative studies. In this paper we present agent-based modelling and simulation (ABMS) as an approach towards this end. An agent-based model of a sociotechnical system describes the performance and interactions of its constituent human operators and technical systems in an operational context. In support of RE it can effectively be used to analyse the capability of a sociotechnical system to deal with disturbances and performance variability. We present an RE cycle, which uses qualitative and quantitative ABMS phases for analysis of the adaptive capacity of a sociotechnical system. The focus in this paper is on the qualitative ABMS phase, including the development of a qualitative model and mental simulation using the qualitative model. The model development is supported by a set of model constructs, which represent key aspects of evolution of agents' states and agents' interactions. The mental simulations use reasoning on the basis of the qualitative model to structurally analyse the interactions and dynamics of the performance in the agent-based model. Results of the qualitative ABMS phase can be used to improve the resilience of operations or they may be followed by quantitative ABMS. The approach is presented in detail for aircraft runway approach operations using conventional systems and an advanced aircraft surveillance application system.

© 2016 Elsevier Ltd. All rights reserved.

1. Introduction

1.1. Resilience engineering

Following the origins of the resilience perspective in ecological studies on prey-predator populations (Holling, 1973), the resilience concept has been adopted in a large number of domains. Various review studies (Francis, 2013; Francis and Bekera, 2014; Hosseini et al., 2016; Martin-Breen and Anderies, 2011) discuss resilience in domains such as ecosystems, socio-ecological systems, socio-economic systems, institutions and governance, social innovation, climate, economy, individual trauma response, psychology, psychiatry, infrastructure, safety management, and organizational science. The resilience concept was introduced in the safety science domain by Hollnagel et al. (2006). For this they coined the term resilience engineering (RE), indicating the ability of a sociotechnical system to adjust its functioning to sustain required operations notwithstanding changes and disturbances,

and the 'engineering' of the sociotechnical system to achieve such ability. RE stresses the key role of performance variability by human operators to adjust for changing demands and conditions in the working context. Safety management that uses an RE perspective leads to what Hollnagel (2014) calls Safety-II, entailing a focus that includes everyday actions and outcomes, which can be contrasted with a Safety-I focus on accidents and incidents only. Bergström et al. (2015) studied a selection of 86 peer-reviewed safety-oriented resilience papers along three questions: why do we need resilience, what is resilience, and who realises resilience? It was found that the need for resilience is typically addressed by referring to the complexity of modern sociotechnical systems and their inherent risks. The object of resilience is the capacity to adapt, so as to keep the complex and inherently risky system within its functional limits. The subject of resilience typically is the individual, either at the sharp end or at higher managerial levels.

In a recent RE perspective paper, Woods (2015) discusses four concepts of resilience:

(1) Resilience as rebound, expressing how a system rebounds from disrupting or traumatic events and returns to previous or normal activities.





CrossMark

^{*} Corresponding author.

E-mail addresses: sybert.stroeve@nlr.nl (S.H. Stroeve), mariken.everdij@nlr.nl (M.H.C. Everdij).

- (2) Resilience as robustness, expressing the ability of a system to absorb perturbations.
- (3) Resilience as graceful extensibility, expressing how a system extends performance when surprise events challenge its boundaries.
- (4) Resilience as sustained adaptability, expressing the ability of a system to adapt to future surprises as conditions continue to evolve.

Woods argues that the rebound concept as such provides limited added value, since it needs to be understood what produces a better rebound. For this it needs to be known first what capacities are present before a surprise event arises and how such a surprise event challenges the base capabilities of the system. Woods argues that this implies a shift in focus from the rebound concept to the graceful extensibility and sustained adaptability concepts. With respect to the robustness concept. Woods refers to robust control engineering and indicates that robustness considers a particular system property that is able to withstand a particular perturbation in some sense. As argued above, system brittleness arises when the set of disturbances is not in the system's base capabilities, setting a need for resilience as graceful extensibility and sustained adaptability. In addition, Woods argues that systems that become more optimal in responding to some disturbances tend to become more brittle to other disturbances, addressing the need for system architectures that can sustain the ability to future surprises. In support of graceful extensibility, indicators of system decompensation should be tracked and anticipation of bottlenecks ahead should be stimulated. Sustained adaptability is supported by understanding the effects of changes in a system's life cycle and providing sufficient flexibility to continue to adapt over such longer time scales. In conclusion, a main principle is that a resilient system should be able to well handle surprise events that are outside its design base. How such ability can be achieved is still largely a research subject and new methods are needed for analysis and engineering towards such resilience.

1.2. Modelling for resilience engineering

As ways to assess resilience in various domains, qualitative and quantitative approaches can be distinguished, following a review in (Hosseini et al., 2016). The qualitative approaches include conceptual frameworks and semi-quantitative indices. The conceptual frameworks provide guidelines and best practices for studying resilience in various domains. The semi-quantitative indices are based on expert assessments of different qualitative aspects of resilience, for instance by structured sets of questions that are scored on a Likert scale. The quantitative approaches include general measures for resilience quantification and domain-specific structural-based modelling approaches. As general measures, a broad range of deterministic and stochastic measures are presented in Hosseini et al. (2016), which all somehow describe the decline and recovery of system performance following a disturbance. The structural-based models include optimization models, simulation models and fuzzy logic models, which mostly describe the vulnerability and recovery for disturbances in networks (e.g. transportation, power transmission, communication) and supply chains. Such models tend to describe system performance at relatively high and aggregated system levels, such as network nodes and average consumption, rather than at the level of interacting humans and technical systems in a sociotechnical system. As such they remain at a distance from RE needs.

In RE, typically qualitative approaches are used to assess resilience, and to improve resilience on the basis of such understanding. The results of such studies include guidelines for performing resilience research, qualitative insights into safety occurrences, or

qualitative recommendations for design. Typically these studies discuss sociotechnical systems in detail, including interacting humans and technical systems. A well-known qualitative approach is the Functional Resonance Analysis Method (FRAM) developed by Hollnagel (2012). It uses a functional analysis-based approach, wherein functions (e.g. activities, tasks) in an operation are described by six aspects, performance variability of functions is identified, relations between functions and propagation of performance variability that may lead to functional resonance is analysed, and these analysis results are linked to the consequences for the operation. FRAM has been applied for retrospective analysis of incidents and accidents (Herrera and Woltjer, 2010; Paulo Victor Rodrigues, 2011) as well as for prospective analysis in system design (Macchi et al., 2011; Praetorius et al., 2015). It is recognized in Praetorius et al. (2015) that notwithstanding the potential of FRAM to uncover operational complexity given particular events, it is difficult to analyse and model everyday operations that do not include such events, and it may be hard to convey field data into functional models. Also other RE studies often use incidents, accidents or some kinds of non-nominal events as basis for their analysis. Thus, it is often still hard to use RE approaches productively for understanding everyday actions and outcomes, such as advocated in the Safety-II perspective of Hollnagel (2014).

We argue that there is a need in RE for more structured modelling approaches for analysis of resilience in sociotechnical systems that can support qualitative as well as quantitative studies. Support of qualitative studies is needed to align with customary approaches for studying sociotechnical systems and with the vocabulary of their practitioners, such that multidisciplinary contributions to the analysis can be achieved. Furthermore, there are cases in which a qualitative study provides sufficient results and no further detailing towards quantification is needed. Prime reasons for structured modelling and quantification in RE are to better understand complex sociotechnical systems' behaviour, and to develop more specific design requirements. Relations, events and dynamics of sociotechnical systems can be manifold and they can be hard to analyse and understand without structured means. Modelling and simulation provide such structured means for attaining deepened understanding of sociotechnical systems. Given the key contributions of human behaviour and performance variability for resilience (Hollnagel et al., 2006), it is essential that human roles and performance variability are well represented in such modelling and simulation.

1.3. Agent-based modelling and simulation

In this paper we present agent-based modelling and simulation (ABMS) as a structured approach for RE of sociotechnical systems. ABMS is an approach for modelling complex systems by describing the behaviour and interactions of a collection of autonomous decision-making entities, called agents (Bonabeau, 2002; Macal and North, 2010; Van Dam et al., 2013). The overall system behaviour emerges as a result of the individual agent processes and their interactions. ABMS provides a highly modular and transparent way of structuring a model, thus supporting systematic analysis, both conceptually and computationally. ABMS has been used in a wide range of application fields, including molecular physics, cell biology, ecology, epidemiology, social sciences, economy, market analysis, archaeology, anthropology, and transport and traffic (Chen and Cheng, 2010; Macal and North, 2010). In safety studies, ABMS has been used for accident risk assessments (Blom and Bakker, 2012; Everdij et al., 2014; Stroeve et al., 2013a).

An agent-based model of a sociotechnical system describes the performance and interactions of its constituent human operators and technical systems working in an operational context. In studying resilience of a sociotechnical system it is key to understand the system's capability to deal with disturbances and performance variability. Such disturbances and performance variability may reflect a wide range of events, conditions or circumstances, and they may be internal to the sociotechnical system, i.e. stem from particular human operators or technical systems, or they may be external, i.e. reflect phenomena in the environment of the sociotechnical system. Human operators and technical systems can express a large variety of behavioural patterns, which are influenced by processes and characteristics of the agent considered (e.g. cognitive and affective aspects), and which depend on interactions between the agents. ABMS offers the possibility to combine a large variety of models for expressing the behaviour and performance variability of the interacting agents in a sociotechnical system. Simulations for ranges of varying conditions can help to unravel the complexity of the sociotechnical system, and to find ways to effectively support its adaptive capacity. As such ABMS is a very promising technique in support of RE.

1.4. Agent-based modelling and mental simulation

We present ABMS as part of a generic cycle for RE, wherein we distinguish two ABMS phases: qualitative ABMS and quantitative ABMS. Qualitative ABMS includes the development of a qualitative agent-based model and it uses this model for reasoning on relations and dynamics of agents' states; this use of the model we call "mental simulation". Quantitative ABMS includes development of a formal model, software implementation and computer simulation; this is the customary type of simulation in ABMS research. In this paper we focus on the principles of ABMS for RE and on the qualitative modelling phase, including mental simulation for evaluation of the model.

Mental simulation is a well-known concept in neuroscience and psychology, which describes the act of imagination and the generation of alternative realities, and which is used for explaining a broad variety of phenomena such as affect, motivation, behaviour and motor control (Markman et al., 2012). This process of selfprojection into alternate temporal, spatial, social, or hypothetical realities is considered a distinctively human capacity, which provides meaning in life (Waytz et al., 2015) and which allows humans to participate in a complex sociocultural world (Baumeister and Masicampo, 2010). Mental simulations are also used in engineering and science as ways to reason about interacting and evolving processes, e.g. for mechanical systems (Hegarty, 2004), or for generating new scientific hypotheses and theories (Clement, 2008; Nersessian, 1999). In particular, mental simulation is the cognitive process for building and interpreting thought experiments (Nersessian, 1992), which are described as "... the construction of a dynamical model in the mind by the scientist who imagines a sequence of events and processes and infers outcomes. She then constructs a narrative to describe the setting and sequence in order to communicate the experiment to others...". Such a narrative basically is a linguistic description of the spatial, temporal and causal relationships among events and entities, but it can be enhanced by supporting material, such as diagrams, pictures and maps. Nersessian (1992) discusses experimental evidence which indicates that such model-based reasoning is faster than reasoning with (logical) propositions, as well as evidence that a reader of a narrative spontaneously constructs mental models to represent and reason about the situations depicted by the narrative. In later work (Chandrasekharan et al., 2013), thought experiments are considered to be on a spectrum of simulative model-based reasoning, together with physical models and computational models. Herein it is argued that computational models can provide deeper insights than thought experiments, since they better support examining a wide range of possibilities within the model parameter space. An agent-based model provides a detailed and structured representation of a sociotechnical system, which ultimately is on the computational side of the spectrum for simulative model-based reasoning. During the development of an agent-based model, it is customary to use mental simulations to arrive at narratives on the ways that agents perform and interact, and to reason whether all essential entities and types of performance are represented in the model for the research question posed.

1.5. Objective and structure of the paper

It is the objective of this paper to show that agent-based modelling and simulation is a suitable modelling approach for resilience engineering, and that mental simulation on the basis of a qualitative agent-based model effectively supports attaining insights in the dynamic relations and performance of a sociotechnical system. The approach is illustrated in detail by an air traffic application case for aircraft approach operations towards a runway using conventional systems and an advanced aircraft surveillance application system (ASAS).

Section 2 presents the ABMS-supported RE cycle and it describes the methods for the development of a qualitative agent-based model and for the mental simulations using this model. Section 3 introduces the RE study of the air traffic application case. Section 4 describes the development of a qualitative agent-based model for these air traffic operations, which addresses the performance of interacting human and technical system agents. Section 5 describes the results attained by mental simulation of the qualitative agent-based model for agents' interactions and evolution of agents' states, and implications of these results. Section 6 provides a discussion of the ABMS approach for RE.

2. ABMS approach for resilience engineering

2.1. ABMS-supported resilience engineering cycle

In this paper, ABMS is part of an RE approach (Pinska-Chauvin et al., 2016), which includes the following main steps (Fig. 1):

Step 0. *Scope RE study:* to determine the objectives and scope of the RE study;

Step 1. *Describe operations:* to describe the sociotechnical system and its operations in nominal conditions;

Step 2. *Identify varying conditions*: to identify all kinds of disturbances and performance variability that can influence operations by the sociotechnical system;

Step 3. *Analyse adaptive capacity:* to identify and understand the strategies applied by the sociotechnical system for dealing with the varying conditions;

Step 4. *Improve resilience:* to identify means to improve the resilience of the sociotechnical system. Such changes in the operations may induce a need for another round in the RE cycle.

Prime methods in these steps are workshops with human operators who are experienced in the operations studied. Such workshops contain specific sessions for each of Steps 1–4 (Everdij et al., 2016).

ABMS contributes to the analysis of adaptive capacity in Step 3 of this RE cycle. As a first sub-step (3a) strategies that are applied for dealing with varying conditions are identified and discussed in a workshop with operators. This provides an initial analysis of the adaptive capacity of the sociotechnical system for dealing with the identified varying conditions in the context of the operations. In some cases, such analysis may be sufficient to gain a proper understanding of the adaptive capacity, such that this understanding



Fig. 1. Steps in an ABMS supported resilience engineering cycle.

suffices for the identification of measures to improve the resilience. In other cases it may be recognized that the complexity of the interrelations and dynamics of the sociotechnical system is too high to attain a sufficient understanding of the adaptive capacity. In these cases ABMS provides a structured means to obtain a more profound understanding of the adaptive capacity of the sociotechnical system. In support of Step 3, we distinguish two phases in ABMS: (3b) qualitative ABMS and (3c) quantitative ABMS.

The qualitative ABMS phase is the first phase, which always precedes the quantitative ABMS phase, but which can also be the sole result of ABMS in support of Step 3. It consists of the following steps:

- Development of a qualitative model. The scope of the model development is determined. This scope may focus on particular (more complex) parts of the operation in relation with particular varying conditions. Next the agents and interactions between the agents are determined. For each of the agents it is determined what model constructs are needed to describe the states and behaviour of the agent and details of the model constructs are provided qualitatively.
- *Mental simulation.* On the basis of the developed qualitative model it is reasoned by the model developers what kinds of sequences of agents' interactions exist and in what way performance variables of the model may develop given particular varying conditions. Such mental simulation can serve to provide feedback in the model development process by identifying weak spots in the developed model. The mental simulation also can lead to new insights into the coupling and dynamics of the agents as a basis for recommendations to improve resilience of the sociotechnical system.

The quantitative ABMS phase builds upon the qualitative ABMS phase and it consists of the following steps:

- *Development of a formal model.* The qualitative agent-based model is further formalized to arrive at a complete mathematical description of the agent-based model. Parameters in the model are provided with quantitative values. Such parameter values may be uncertain and it can be useful to specify uncertainty bounds.
- *Software implementation.* The formal model is used to develop computer simulation code of the agent-based model, e.g. using general programming languages or a specific agent-based modelling tool (Macal and North, 2010). The performance of the software is verified.
- Computer simulation. The software implementation is used for computer simulation of the agent-based model and the simulation results are interpreted by the user. The computer simulation results provide detailed insights into the couplings, dynamics and variability of the agents' performance. The insights thus achieved can be the basis for qualitative or quantitative recommendations to improve the resilience of the sociotechnical system.

The focus of this paper is on the methods and results of the (3b) qualitative ABMS phase and the following two subsections describe the methods of the development of a qualitative model and of the mental simulation using this qualitative model.

2.2. Development of a qualitative agent-based model

2.2.1. Introduction

An agent-based model of a sociotechnical system describes the performance and interactions of its constituent human operators and technical systems working in an operational context. Agents in a sociotechnical system contain boundaries separating internal states and processes from states and processes external to the agent (in other agents/environment). Relations between an agent's internal and external states or processes are represented strictly via the inputs and outputs of the agent considered. This makes it easier to specify models of complex systems that consist of many interacting entities, thereby facilitating effective study of the emergent behaviour of such systems. The particular ways that an agent's states evolve, the implications of an agent's input, and the behavioural patterns and output of an agent can be represented by model constructs. Also events and conditions that make up the environment of the sociotechnical system can be represented by model constructs (see Fig. 2).

The development of a qualitative agent-based model of a sociotechnical system for RE is done along the following steps:

- Scoping of the qualitative agent-based modelling;
- Identification of agents and interactions;
- Identification of model constructs;
- Qualitative description of model details.

These steps are explained next.

2.2.2. Scoping

The scope of the qualitative agent-based modelling refers to the types of operations, the human operators and technical systems, the varying conditions, and the geographical boundaries of the sociotechnical system. Typically, the scope of the qualitative agent-based modelling is a subset of the scope of the overall RE study (as described in Step 1 of Fig. 1). In particular, the scope includes those aspects of the performance of the sociotechnical system for which the results of the analysis in Step 3a are too uncertain and for which it is expected that additional analysis by ABMS can reduce the level of uncertainty.

2.2.3. Identification of agents and interactions

The sociotechnical system in the scope of the study is modelled by a set of agents. In Macal and North (2010) essential characteristics of an agent are considered to be that an agent is a selfcontained uniquely identifiable individual, that it is autonomous and can function independently, that it has a dynamic state, and that it has dynamic interactions with other agents that influence its behaviour. In the modelling of a sociotechnical system the agents typically are human operators and technical systems. To restrict the complexity of the overall agent-based model, a group of human operators or a group of technical systems may be considered as a single (aggregated) agent, if it can be considered as a single entity in its behaviour and interactions with other agents. In this modelling step, it is decided what the agents of the studied sociotechnical system are, and which inter-agent interactions exist. This builds upon the understanding of the operations in nominal situations achieved in Step 1, as well as on the knowledge of strategies for dealing with particular varying conditions achieved in Step 3 of the RE cycle.

2.2.4. Identification of model constructs

A model construct is a generic model describing particular aspects of the ways that agents behave and evolve in interactions with other agents and conditions in the environment. Model constructs are also used to describe evolutions in the environment of the agents. Model constructs are high-level archetypes and within specific applications, modelling details need to be specified at a later stage (see Section 2.2.5).

For the identification of model constructs it needs to be understood, which key aspects of the entities in the sociotechnical system drive their behaviour and contribute to the uncertainty in the overall performance, for the conditions in the scope of the ABMS study. Such understanding can be achieved in the initial analysis of the adaptive capacity of Step 3a and this also forms the basis for the scoping of the ABMS study.

Next, this knowledge is used to identify the set of model constructs that can best represent the key behavioural aspects of the agents in the model. This identification requires an understanding of types of agent-based models, their background, and the ways that they are applied in ABMS. Notwithstanding the systematic background knowledge that is used, to a certain extent the



Fig. 2. Generic overview of an agent-based model, consisting of human operators and technical systems in an environment. Agents are represented by a combination of model constructs that represent aspects of the agents' behaviour and interactions.

identification of model constructs is also a creative process, which builds upon the experience and preferences of the modeller. As such, there is no unique set of model constructs for a particular problem. Also, if no suitable model construct exists to represent a particular aspect of the sociotechnical system, a new model construct can be developed to this end.

As a basis for the identification of model constructs, the ABMS literature provides a large variety of agent models (Macal and North, 2010), including for modelling of sociotechnical systems (Van Dam et al., 2013) and social interactions (Sun, 2006). In support of RE in air traffic management (ATM), Stroeve et al. (2013b) developed a library of model constructs for agent-based modelling. This set contains 38 model constructs, which were identified in the ABMS literature and which were evaluated for their capability to support the modelling of a broad range of conditions and events that may contribute to unsafe situations. These model constructs include quite generic ones, e.g. describing how situation awareness may evolve or generic tasks are scheduled and performed, as well as quite detailed and specific ones, e.g. about confusion, group emotion, and goal-oriented attention. For studying resilience in air transport, this library provides a useful starting point for the identification of model constructs.

2.2.5. Qualitative description of model details

In this step, the details of the model constructs are determined at a qualitative level for all agents in the use case. The specifically required model details depend on the model construct considered. The list below provides an overview of the types of details that may be specified in this stage.

- State variables, such as the position and speed of an aircraft, or the situation awareness of an air traffic controller about the position of an aircraft;
- Mode variables, describing an operating mode of a technical system (e.g. some normal working mode, or a failure mode), or of an human operator (e.g. tactical or opportunistic contextual control mode);
- Types of tasks that a human operator may perform;
- Types of behaviour that an agent may show;
- The way that a model construct is influenced by other model constructs within the same agent (intra-agent input);
- The way that a model construct influences other model constructs within the same agent (intra-agent output);
- The way that a model construct is influenced by model constructs of other agents (inter-agent input);
- The way that a model construct influences model constructs of other agents (inter-agent output).

All these aspects are provided qualitatively using textual descriptions. Quantification of the agent-based model is not done in this step, but only in Step 3c of the RE cycle (not detailed in this paper).

2.3. Mental simulation

The developed qualitative agent-based model provides a structured representation of the sociotechnical system for a particular operational context. Such structured representation helps to improve the understanding of the complexity of the relations and dynamics of the sociotechnical system. The development of the qualitative model as such is a first way to support this understanding. As a second and more illustrative way, mental simulation employs reasoning on the basis of the developed qualitative agent-based model. We next discuss two types of mental simulation, which focus on the interactions in the agent-based model and on its dynamics, and we discuss the use of the mental simulation results in the RE cycle.

2.3.1. Analysis of interactions

As a first type of mental simulation, the qualitative agent-based model can be used for structured reasoning about the interactions between the agents' models in the contextual conditions of the operation that are in the scope of the qualitative ABMS phase. Such analysis can focus on interactions during operations that are considered to be normal, or on interactions following some varying condition of interest. It consists of the followings steps.

- As a starting point, an initial condition of agents should be formulated. This initial condition specifies the states and modes of the agents at the start of the mental simulation.
- One or several triggering events or occurrences of varying conditions are specified, which describe conditions of interest for studying the adaptive capacity of the sociotechnical system. If such event/condition occurs at the start of the simulations, it extends the initial condition, otherwise its timing during the scenario is specified.
- Next it is argued what the main changes are in the states and modes of the agents' models following the initial condition. This argumentation is structured by listing sequences of triggers and resulting actions in the agents. Such trigger-action pairs can be internal to an agent (e.g. an observation leading to a decision to coordinate) or it can impose an interaction between agents (e.g. a communication act leading to a change in situation awareness). As such this exercise provides instantiations of sequences of interactions that may occur in the agent-based model. As the state space of the overall model can be extensive, this argumentation is done for the states that are judged to be most relevant for the situation studied. By making such judgments during the simulations, mental simulations differentiate from computer simulations, wherein the overall state space is simulated and wherein only after the simulations the most relevant state transitions are identified.
- A case of multiple varying conditions leads to multiple instantiations of interaction sequences that need to be accounted for, e.g. a bad weather condition versus a bad weather condition in combination with a technical failure.

The ways that the results of such mental simulation can be used in the RE cycle is discussed in Section 2.3.3 and a detailed illustration for the air traffic application is presented in Section 5.1.

2.3.2. Analysis of dynamics

As a next type of mental simulation, the qualitative agent-based model supports structured reasoning about dynamic relations between states and modes of agents. This can most effectively be done in follow-up to the analysis of interactions as explained above, since the types of interactions need to be understood in order to reason about the dynamic effects. It consists of the following steps.

• As a starting point of the analysis it is decided what states or modes need to be studied in detail to support the understanding of the adaptive capacity. These may be key states identified in the analysis of interactions (Section 2.3.1) or other relevant indicators of the overall system performance. It can be especially useful to reason about some aggregated values of states, such as averages of states over ranges of instantiations of the agent-based model.

- An initial condition of the agents' states and modes is specified.
- One or several triggering events or occurrences of varying conditions are specified, which extend the initial condition or occur at a later stage.
- It is qualitatively argued how the relevant states change in time due to the interactions in the agent-based model. The results of this reasoning about the agent states are described in narratives and can be illustrated by graphs as function of time. These graphs provide qualitative indications of the variation in the selected (aggregate) state variables, which are supported by the argumentation of the elements in the agent-based model that are expected to give rise to them. It can be useful to compare the qualitative graphs for several cases, e.g. a new versus an old operation, or an operation in condition 1 versus condition 2. Also in this type of mental simulation, the dynamics of the complete state space are not described in detail, but rather it is judged during the mental simulation what the most relevant state dynamics are.
- In the case of multiple varying conditions, above process needs to account for the triggers they induce for the state dynamics.

The use of this type of mental simulations in the RE cycle is discussed next and a detailed illustration for the air traffic application is presented in Section 5.2.

2.3.3. Use of mental simulation results in the RE cycle

Performing mental simulation requires a good overview and understanding of the qualitative agent-based model, as well as sufficient knowledge about the application field. In addition, experience in agent-based computer simulations or multidisciplinary systems engineering supports meaningful reasoning about the qualitative agent-based model. These requirements typically imply that the prime executors of mental simulations are scientists of the multidisciplinary model design team. Nevertheless, also other independent researchers with a suitable background should be able to understand the model and to execute the mental simulations. This would require some additional effort to understand the overall model, but has the added value of an independent view and interpretation of the model.

Results of the mental simulations can be discussed with operational experts, for instance with the experts who contributed to the identification and discussion of strategies in Step 3a of the RE cycle. The objective of such discussion in Step 3 of the RE cycle is to get feedback from the operational experts on their view of the validity of the mental simulation results. Do they agree with the presented behaviour and interactions of the agents? Do they think that particular key aspects or interactions are missing in the simulation results? Do they understand and agree with the resulting dynamics? Feedback to these types of questions can lead to several types of conclusions in the context of the RE cycle:

- It can support the identification of missing aspects in the agentbased model, e.g. particular agents, behavioural aspects or agent interactions. This may require an update of the qualitative agent-based model and of the mental simulations, followed by another feedback round of operational experts.
- It can lead to the conclusion that all key elements of the sociotechnical system are well represented in the agent-based model, but that the methods of mental simulation are not sufficiently sophisticated to understand the dynamic and stochastic implications of the agents' behaviour and interactions with sufficient certainty. Following this conclusion, it can be decided to further develop the agent-based model towards a formally specified and quantified model, which is evaluated using computer simulations in Step 3c of the RE cycle.

• It can lead to the conclusion that the mental simulations provided sufficient insights into the adaptive capacity. The thus achieved insights can be used in Step 4 of the RE cycle as a basis to identify measures to improve the resilience of the sociotechnical system.

3. Studying the resilience of an air traffic operation

3.1. Air traffic operation

The air traffic application case of this paper considers the approach of multiple traffic streams towards a single runway. A schematic overview of the air space organization is shown in Fig. 3. Aircraft enter via sectors S1 or S2, heading towards waypoints W1 or W2, respectively. The two traffic streams merge at W3 in the ARR (Arrival) sector. The last part of the approach and landing on the runway is controlled in the TWR (Tower) sector. Each sector is under the control of a single air traffic controller, who is responsible for safe and efficient traffic through the sector. Prime means for this are radar surveillance systems and radioenabled voice communication between pilots and controllers. In conventional air traffic control, pilots are given instructions to adhere to particular speeds or to use series of heading directions (vectoring) such that separation between aircraft is maintained in the sequence and during merging. There is a joint supervisor for the controllers working in sectors S1, S2 and ARR, and there is a supervisor for the TWR sector.

The future operation studied uses an aircraft surveillance applications system (ASAS) to enable airborne spacing (ASPA) for sequencing and merging. Pilots can get an instruction by the S1, S2 or ARR controller to use ASAS to maintain a particular spacing with respect to a target aircraft in front of them. If the target aircraft is on the same route, ASAS is used for sequencing only (e.g. aircraft 3 uses aircraft 2 as a target for sequencing in Fig. 3). If the target is on the merging route, ASAS is used for merging first and sequencing next (e.g. aircraft 4 uses aircraft 3 as a target for merging and sequencing in Fig. 3). Following the controller instruction the pilots first use ASAS to find out whether the requested airborne spacing is acceptable, and if so, they affirm the controller's request and initiate the airborne spacing operation. Now ASAS continuously tracks the spacing with respect to the target aircraft and it adjusts the aircraft speed (within limits) to attain and maintain the requested spacing. Although the spacing is maintained by ASAS under the pilots' supervision, the controller is still the person responsible for guarding the separation minima.

3.2. Workshop-based resilience analysis

The operations outlined in the previous section were studied for conventional and ASAS approach operations at a large European airport. Resilience of these operations was studied by a workshop-based analysis following a first cycle of Steps 0-4 as described in Section 2.1. As such, workshop sessions with air traffic controllers and airline pilots were used as the prime means to describe current and future operations, identify varying conditions, identify strategies to deal with the varying conditions, and to identify means to improve the resilience of the sociotechnical system (Everdij et al., 2016). This approach led to qualitative interpretations of the adaptive capacity by structuring of the workshop discussions with the operational experts. It was recognized in the study that the analysis of the adaptive capacity may benefit from additional more detailed analysis approaches, such as agentbased modelling and simulation. The development of a qualitative agent-based model for these operations and results of mental simulation for these models are presented next in Sections 4 and 5, respectively.



Fig. 3. Schematic top view of air traffic routes approaching a runway with a sample of aircraft.

4. Development of a qualitative agent-based model

This section shows the development of a qualitative agentbased model along the steps of Section 2.2 for the operation described in Section 3. Section 4.1 describes the scope of the agent-based modelling. Section 4.2 provides an overview of the agents and their interactions. Section 4.3 presents the selection of the model constructs. Section 4.4 presents the models of the human agents. Section 4.5 presents the models of the technical system agents. Section 4.6 presents the models of the environment of the sociotechnical system.

4.1. Scope

It was decided to include the conventional approach operation as well as the future ASAS approach operation in the scope of the agent-based modelling study. This provides the opportunity to compare results of both cases. With respect to the human operators, all air traffic controllers, supervisors and airline pilots operating in the airspace sectors, as discussed in Section 3.1, are included in the scope. For the technical systems, the air traffic control (ATC) communication systems, ATC surveillance systems, the communi cation-navigation-surveillance (CNS) systems of the aircraft, the ASAS systems of the aircraft, and the aircraft as flying entities themselves are in the modelling scope. The geographical boundaries of the model reflect the aircraft entry in sectors S1 or S2 and the aircraft exit when they pass the runway threshold. The varying condition in the scope of the modelling is a situation of sudden and unexpected bad weather at the airport, which deteriorates the runway condition and therefore leads to the need to reduce the runway capacity.

4.2. Agents and interactions

Fig. 4 shows an overview of the identified agents and interactions in the agent-based models of the conventional and ASAS cases. It represents a sequence of aircraft and the air traffic control provided by the controllers during the aircraft approaches. Within each aircraft we distinguish the pilots as an aggregated agent, the aircraft flight performance (position, speed, heading, etc.), the aircraft CNS system, and the ASAS system. The controllers use ATC communication systems and ATC surveillance systems as prime technical systems. Supervisors in the tower and S1/S2/ARR control room decide on modes of operations and interact with the controllers. The integrated set of these agents is considered as the sociotechnical system. It exists in an environment, consisting of airspace and weather in the current model instantiation.

4.3. Selection of model constructs

As a basis for the identification of model constructs for the air traffic application case we used a library of 38 agent-based model constructs that was developed in support of RE in ATM (Stroeve et al., 2013b). For each of these model constructs we described its potential role in an agent-based model for the use case and we assessed whether the model construct is needed. A leading argument in this assessment was that in this development stage the overall agent-based model should not be overly complex, but the main interactions and behavioural aspects of the sociotechnical system should be represented.

As a result of this assessment, a set of 11 model constructs was chosen for modelling of the application case. These model constructs are introduced is the list below. More details on the way that they are applied and integrated in the use case are provided in Sections 4.4–4.6.

- *M1: Multi-agent situation awareness.* Situation awareness (SA) is a well-known and much discussed human factors concept for the perception of elements in the environment, their interpretation and the projection of the future status (Endsley, 1995). In agent-based modelling, ascription of mental qualities (e.g. beliefs, desires) to technical systems is seen as useful for analysis of complex systems (Wooldridge and Jennings, 1995). For analysis of safety risks in complex sociotechnical systems, Stroeve et al. (2003) developed the multi-agent SA model construct to describe the development of states of humans and technical systems with regard to their perception, interpretation and future projection of their multi-agent environment. Observation, communication and reasoning processes of the agents drive such state development.
- *M2: Task identification.* This describes the ways that the operator identifies the tasks that need to be performed at a particular time instance.



Fig. 4. Interactions in the agent-based models for the conventional case (without ASAS elements) and for the ASAS case (all elements).

- *M3: Task scheduling.* This describes which tasks may be performed concurrently, as well as priority among the tasks that cannot be performed concurrently (Blom et al., 2001).
- *M4: Task execution.* This describes the performance of a human operator with regard to the execution of a specific task, including task-specific performance characteristics.
- *M5: Contextual control mode.* This model construct considers that humans can function in a number of contextual control modes, such as Strategic, Tactical, Opportunistic and Scrambled (Hollnagel, 1993). The contextual control mode may depend on human performance aspects such as the range of tasks to be done and the situation awareness of the human. It influences human performance aspects such as the planning horizon and the accuracy of task performance.
- *M6: Task load.* This model construct describes the number and complexity of tasks that need to be performed, as considered in the task scheduling process. The task load influences the

contextual control mode of the human operator. At a more detailed level, the task load may also describe the resources required by tasks at the level of visual, auditory, cognitive and motor performance.

- *M7: Decision making.* This model construct describes decision making by agents on the basis of the situation awareness and decision rules or processes.
- *M8: System mode.* This model construct describes the behaviour of a technical system by different modes. These modes are discrete states for the functioning of technical systems, e.g. failure conditions and system settings. Mode changes may be deterministic or probabilistic and they may depend on the functioning of other agents.
- *M9: Dynamic variability.* This model construct describes the variability of states of agents due to dynamic processes, e.g. the movements of an aircraft according to differential equations relating states such as position, velocity, acceleration and thrust.

- *M10: Stochastic variability.* This model construct describes the stochastic variability in the performance of human operators and technical systems, e.g. variability in task duration or system accuracy. This stochastic variability also includes large deviations from normal or intended practices in human performance, which may be labelled as 'errors' in hindsight (Dekker, 2005). The level of stochastic variability may be influenced by the contextual control mode.
- *M11: Contextual condition.* This model construct describes the context of the operation, such as weather, route structure, environmental conditions and airport infrastructure.

4.4. Human agents models

The human agents in the models are S1 Controller, S2 Controller, ARR Controller, TWR Controller, Supervisor S1/S2/ARR, Supervisor Tower, and Pilots Aircraft. These human agents are all modelled by the same set of model constructs as shown in Fig. 5; these are all the model constructs identified in Section 4.3 except M8 and M11.

Multi-agent situation awareness (M1) is a key modelling construct, which represents the situation awareness of a human agent with respect to other agents and the environment of the agent. The situation awareness is updated by task execution (M4), e.g. communication, observation, or by decision making (M7), and vice versa, task execution and decision making depend on the agent's situation awareness. During task execution, input of other agents or elements in the environment may be obtained and output may be sent to other agents or elements in the environment. The identification and scheduling of tasks' execution is represented by four modelling constructs:

- Task identification (M2), which determines whether a task should be done, triggered by some internal timing mechanism, by the agent's situation awareness, or by an external cue;
- Task scheduling (M3), which determines when a particular task is performed;
- Task load (M6), which represents the load of the identified tasks; and
- Contextual control mode (M5), which represents the control mode of the human agent and which influences the scheduling and execution of tasks.

The remaining model constructs represent the dynamic performance of the agent (dynamic variability, M9), and stochastic/random elements in the behaviour of the agent (stochastic variability, M10).

To provide more insight into the application of the model constructs in the human agents, Tables 1 and 2 present details for the ARR controller agent. Similar models were developed for the other human agents. The basis for the task-related model constructs M2, M3 and M4 are a number of tasks that the ARR controller can perform in the context of the approach operations. Table 1 shows these tasks and for each task it is indicated whether it can be used in the ASAS or conventional cases, what the trigger is for task identification (M2), what the task priority is and whether a task can be



Human operator agent

Fig. 5. Model constructs of a human operator agent and their interactions.

Table 1

Modelled ARR controller tasks and the related model constructs M2, M3 and M4. For each task the following aspects are indicated: whether it can be used in the ASAS and/or conventional case (yes/no); what the trigger is for task identification; what the task priority is (1 = highest priority); whether a task can be done concurrently (yes/no); what is done in task execution.

Task		Case		M2. Task identification	M3. Task scheduling		M4. Task execution
No	Description	ASAS	Conv	Trigger	Prior- ity	Concurrency	
A	Decide on initiation of ASPA operation	У	n	When aircraft has passed a particular point; regular re- evaluation until decision reached	11/12	У	Update SA about initiation of ASPA operation
В	Provide ASPA clearance to aircraft-i: merge or remain behind	У	n	When it was decided in Task A	11/12	n	Communicate ASPA clearance to aircraft-i
С	Communicate termination of ASPA operation to multiple aircraft	У	n	When told so by supervisor (in Task D)	6	n	Communicate termination of ASPA to a sequence of aircraft, starting with the aircraft that is closest to the entry of the ARR sector
D	Coordinate with supervisor about separation and possibly ASPA end	У	У	When initiated by supervisor	5	n	Update SA about separation standard and possibly ASPA termination
E	Handover from S1 or S2 controller	У	У	When initiated by S1 or S2 controller	13/14	n	Communicate with handed over aircraft-i
F	Handover to TWR controller	У	У	When aircraft is 6 NM from runway	13/14	n	Instruct aircraft-i to contact TWR
G	Terminate ASPA operation for single aircraft	у	n	When aircraft reports problem or when separation conflict is detected	3	n	Update SA about termination of ASPA for single aircraft and communicate to aircraft
Н	Monitor traffic situation and spacing	У	У	When aircraft is handed over, regular tracking of each aircraft	4	У	Monitor traffic situation and spacing between aircraft in the sequence
Ι	Decide on aircraft control in normal control situation	У	У	When monitoring (task H) is done	7	У	Update SA about planned control action
J	Decide on intervention (heading, speed and/or altitude) in case of separation conflict	У	У	When separation conflict is detected	1	У	Update SA about planned intervention action
К	Provide navigation clearance (route, waypoints) to aircraft-i	У	У	When it was decided in Task I	8/9/10	n	Communicate navigation clearance to aircraft-i
L	Provide heading, speed or altitude instruction to aircraft-i in normal control situation	У	У	When it was decided in Task I or Task J	8/9/10	n	Communicate heading/speed/altitude to aircraft- i in normal situation
Μ	Provide heading (vector), speed or altitude instruction to aircraft- i in separation conflict	у	У	When it was decided in Task I or Task J	2	n	Communicate heading/speed/altitude to aircraft- i in separation conflict
N	Provide vector back to route to aircraft-i	У	У	When it was decided in Task I	8/9/10	n	Communicate vector back to route to aircraft-i

done concurrently with other tasks (M3), and what the implication of task execution is (M4). For all model constructs, Table 2 shows their main performance aspects (e.g. situation awareness components, contextual control modes, task load influences), their input/output relations within the agent, and their input/output relations with other agents.

In general, the multi-agent situation awareness model describes what kinds of aspects of the overall sociotechnical system each human agent may be aware of, and how this dynamic knowledge state can be influenced by other agents. Application of the taskrelated model constructs is based upon identification of the tasks done by the humans in the operations considered. In the model development, 13-14 tasks were identified for the main operators in the operation (S1, S2 and ARR controllers and pilots), whereas 4-5 tasks were considered sufficient for the supervisors and tower controller models. As next aspects of the task analysis, it was argued what the triggers are for identification of each of the tasks, what the priorities are in the sets of tasks, and what tasks can be done concurrently with other tasks. Finally, it was argued what the impact is of task execution. In a multi-agent environment, task execution often implicates an effect of one agent on another agent, for instance by communication, observation, or setting of technical system parameters. These kinds of actions typically imply updating of situation awareness of agents in the multi-agent environment. Such updated situation awareness components can be, in their turn, triggers for task identification by the agent considered, and they are input for the decision making model construct. By such situation awareness-based interactions between agents, the multi-agent situation awareness construct is key for analysis of information flows in the sociotechnical system and it forms the link with the task-related performance of the human agents. Next, aspects such as number, duration, complexity and frequency of tasks identified by a human agent effectuate the task load. The task load is considered to be a key input parameter of the contextual control mode of a human agent and this control mode has effect on the task scheduling and task execution. Finally, many model constructs, such as task and decision making related constructs, include dynamic and stochastic variability, such as variability in task durations or noise or errors in observations, and these are explicitly represented in the agent-based model.

4.5. Technical system agents models

4.5.1. ASAS agent

The ASAS agent is part of the ASAS case only. Its model constructs are shown in Fig. 6. They represent the situation awareness (M1) of an ASAS agent about the position, speed and type of the ownship and of the target aircraft, the control actions performed by the ASAS system (decision making, M7), and system modes (M8), such as sending and receiving information to the aircraft and pilots. Details of the performance and the input-output relations in the agent-based model are shown in Table 3. Instantiation of model constructs for ARR controller agent: specific aspects, input/output relations within the agent, input/output relations with other agents. Some elements

Table 2

apply to the ASAS case only. M1. Multi-agent situation awareness SA components • General control mode: ASPA on/off *ASAS case only* • List of aircraft under control • ASPA mode of each controlled aircraft-i: on / off *ASAS case only* • 3D position of each controlled aircraft-i Speed of each controlled aircraft-i • Track of each controlled aircraft-i • Instruction/clearance to each controlled aircraft-i: remain behind aircraft-i/merge behind aircraft-i/fly a vector/fly a speed/descend to and/or maintain altitude/fly a published procedure Intra-agent Input • M4 Task execution: executing a task, e.g. a monitoring task, may lead to an update of SA • M7 Decision making: reaching a decision leads to an update of SA • M2 Task identification: recognition based on SA that a task needs to be done Output • M4 Task execution: task execution influenced by SA, e.g. sending a particular message • M7 Decision making: decision making based on SA Inter-agent • ATC Surveillance Systems: aircraft position/speed/track Input • Pilots aircraft-i: ASPA mode *ASAS case only* Supervisor S1/S2/ARR: general control mode ASPA *ASAS case only* Output No direct effects on other agents, since this is always done via the agent's task execution M2. Task identification Task triggers See Table 1 Intra-agent Input • M1 Multi-agent situation awareness: SA of controller can be a trigger for a task • M10 Stochastic variability • M3 Task scheduling: notification that a task needs to be done Output • M6 Task load: keeping track of the number and types of tasks to be done • Supervisor S1/S2/ARR: incoming message Inter-agent Input Pilots aircraft-i: incoming message Output No direct effects on other agents M3. Task scheduling Scheduling aspects See Table 1 Intra-agent Input • M2 Task identification: a task is scheduled after it has been identified M9 Dynamic variability Output • M4 Task execution: task scheduling determines when a task is executed Inter-agent Input None Output No direct effects on other agents M4. Task execution Execution effects See Table 1 • M1 Multi-agent situation awareness: task execution may depend on the situation awareness of the ARR controller Intra-agent Input • M3 Task scheduling: task scheduling determines when a task is executed • M5 Contextual control mode: the control mode may affect the way that a task is performed • M7 Decision making: more complicated tasks may need dedicated decision making M9 Dynamic variability M10 Stochastic variability • M1 Multi-agent situation awareness: situation awareness of controller may be updated following task execution Output • M7 Decision making: decisions made may influence the task execution Inter-agent Input None • Pilots aircraft-i: outgoing messages Output M5. Contextual control mode Control modes Opportunistic Tactical Mode switches • Switch from tactical to opportunistic if task load is higher than a threshold • Switch from opportunistic to tactical if task load is lower than a threshold Intra-agent Input • M6 Task load: the task load influences the control mode • M9 Dynamic variability: the control mode influences task duration Output • M10 Stochastic variability: the control mode influences stochastic variability in task execution Inter-agent Input None Output None M6. Task load Task load influences • Number of tasks to be performed • Frequency of each task to be performed · Expected duration of each task to be performed • Novelty, complexity, difficulty of tasks to be performed Intra-agent Input • M2 Task identification: the number, durations and types of tasks to be done determine the task load Output • M5 Contextual control mode: the task load influences the control mode None Inter-agent Input Output None M7. Decision making Related tasks • Task A: decide on initiating ASPA operation • Task G: decide on terminating ASPA operation • Task I: decide on how to move aircraft safely and efficiently

• Task J: decide on intervention (vector, speed, or altitude to be provided) in case of a separation conflict

S.H. Stroeve, M.H.C. Everdij/Safety Science 93 (2017) 29-49

Intra-agent	Input	 M1 Multi-agent situation awareness: decision making is based on the situation awareness M4 Task execution: decision making may be called upon during task execution
		• M9 Dynamic variability
	Output	• MTO Stochastic variability • M1 Multi-agent situation awareness: decision making undates the controller's situation awareness
	Output	• M4 Task execution: task execution is influenced by the decision making
Inter-agent	Input	None
0	Output	None
M9. Dynamic variabilit	ty	
Examples	-	Duration of task execution
		Duration of decision making
Intra-agent	Input	• M5 Contextual control mode: the control mode can influence the speed of task execution, e.g. higher speed in case of oppor- tunistic control mode
	Output	M3 Task scheduling: dynamics of task scheduling
		M4 Task execution: dynamics of task execution
		M7 Decision making: dynamics of decision making
Inter-agent	Input	None
	Output	None
M10. Stochastic variab	ility	
Examples		Variation in observed spacing during monitoring (Task H)
		Variation in decision on intervention strategy (Task J)
	_	Controller communicates to merge behind wrong aircraft (Task B) *ASAS case only*
Intra-agent	Input	• M5 Contextual control mode: the control mode influences stochastic variability, e.g. a higher probability of a deviation in task execution in the opportunistic control mode
	Output	• M2 Task identification: stochastic variation in task identification
		• M4 Task execution: stochastic variation in task execution
	_	• M7 Decision making: stochastic variation in decision making
Inter-agent	Input	None
	Output	None



Fig. 6. Model constructs of an ASAS agent and their interactions.

4.5.2. Aircraft flight performance agent

The model constructs of an aircraft flight performance agent (Fig. 7) represent the aircraft movements and related aircraft system modes. The dynamic variability construct (M9) represents variations in aircraft movement states (position, speed, heading) and these interact with intra-agent system modes. The system mode construct (M8) represents flight control modes for flying speed profiles, heading profiles, and/or altitude profiles. It has intra-agent interactions with the dynamic variability construct and it has inter-agent interactions with the pilots, ASAS and CNS agents of the own aircraft.

4.5.3. Aircraft CNS agent

The model constructs of the communication, navigation and surveillance (CNS) systems of an aircraft (Fig. 8) represent the agent's situation awareness, communication and other working modes of the CNS system, and the dynamic and stochastic variability in the performance of the CNS system. Situation awareness components in M1 include the 3D positions, speeds and headings of the own and other nearby aircraft, and the identity of these other aircraft. The multi-agent SA construct interacts with system modes in the same agents, which influence the SA updating process, and it interacts with the pilots, ASAS and aircraft flight performance of the own aircraft, with the CNS and aircraft flight performance of other nearby aircraft, and with the ATC communication and surveillance systems. The system modes (M8) represent working modes for aircraft communication, navigation and surveillance systems, including failure modes. They interact with the multi-agent SA construct of the same agent. The dynamic variability construct (M9) signifies the time-dependency in the situation awareness updating processes and the switching in system modes. The stochastic variability construct (M10) signifies random aspects in SA updating (e.g. noise in observations/communications) and in system mode switching.

4.5.4. ATC surveillance system agent

The model constructs of the ATC surveillance system are also represented by Fig. 8. In the context of the ATC surveillance, the multi-agent SA construct (M1) represents the position, speed, heading and identity of the aircraft, the system modes influence the updating process, it receives input from with the aircraft flight performance (primary radar) and CNS system (secondary radar) of the aircraft, and it provides surveillance data to the controllers and the supervisors. The system modes (M8) represent modes of the surveillance system, including failure modes. The dynamic variability construct (M9) signifies the dynamics of the situation awareness updating. The stochastic variability construct (M10) signifies noise in SA updating and random aspects in system mode switching.

4.5.5. ATC Communication agent

The model constructs of the ATC Communication agent (Fig. 9) represent system modes of the ATC communication system, such as the functioning of the voice and datalink communication capabilities, as well as the dynamic and stochastic variability of communication message handling. The interactions are with the controller agents and the CNS systems of the aircraft.

4.6. Environment models

Elements of the environment of the agents in the sociotechnical system are represented by the model construct contextual

Table 3

Instantiation of mod	lel constructs for	ASAS agent: specific aspects, input/output relations within the agent, input/output relations with other agents.
M1. Multi-agent s	ituation awarene	SS
SA components		Control mode: ASPA on/off of own aircraft-i
		• 3D position and track of own aircraft-i
		Speed vector of own aircraft-i
		Type of own aircraft-i
		3D position and track of target aircraft
		Speed of target aircraft
		Type of target aircraft
		• ASPA manoeuvre parameters (target aircraft identity, target aircraft intended flight path information, ASPA manoeuvre type,
		assigned spacing goal, achieve-by point, planned termination point)
		• ASPA execution parameters (interval management speed, predicted spacing interval, measured spacing interval, unable to
		continue and failure flags from the ASAS system)
Intra-agent	Input	• M7 Decision making: reaching a decision leads to an update of the SA
		M8 System mode: reception of messages
	Output	• M7 Decision making: reaching a decision leads to an update of the SA
		M8 System mode: sending of messages
Inter-agent	Input	None
	Output	No direct effects on other agents, as messages are received and sent via system modes.
M7. Decision mak	ing	
Types of decision	IS	Deciding/advising whether ASPA operation can be initiated/terminated
51		• Control algorithm generating interval management speeds to achieve and/or maintain required spacing with target aircraft
Intra-agent	Input	• M1 Multi-agent situation awareness: decision making is based on the situation awareness
0	ľ	M8 System mode: reception of messages
	Output	M1 Multi-agent situation awareness: decision making updates system's situation awareness
	-	M8 System mode: sending of messages
Inter-agent	Input	None
	Output	None
M8 System mode		
Types of system	modes	• Sand interval management speed to aircraft speed control system
Types of system	moues	 Solid interval management spect to ancrait spect of system Receive aircraft data: receive data from own aircraft and target aircraft (via CNS System)
		Receive ASPA mannetwer parameters from nitot
		Provide ASPA manoeuvre and execution narameters for display to pilot
Intra-agent	Input	M1 Multi-agent situation awareness
intru ugent	mput	• M7 Decision making
	Output	 M1 Multi-agent situation awareness: decision making undates system's situation awareness
	output	M8 System mode: sending of messages
Inter-agent	Input	• CNS of own aircraft-i
		Pilots of own aircraft-i
		Aircraft evolution of own aircraft-i
	Output	CNS of own aircraft-i
		Pilots of own aircraft-i
		Aircraft evolution of own aircraft-i



Fig. 7. Model constructs of an aircraft flight performance agent and their interactions.



Fig. 8. Model constructs of Aircraft CNS System agent, as well as of ATC Surveillance System agent, and their interactions.



Fig. 9. Model constructs of an ATC communication agent and their interactions.

condition (M11, Fig. 10). In the current model instantiation, simple models are used for the airspace and the weather. The airspace model represents the fixed structure of the airspace, such as the locations of waypoints and of the runway (see Fig. 3). The weather model represents the weather condition at the airport by two modes for good weather and bad weather, and it uses stochastic switches between these modes.

5. Mental simulation of the agent-based model

The qualitative agent-based model presented in Section 4 is used for mental simulation according to the approach described



Fig. 10. Model construct of environment.

in Section 2.3. The mental simulations have been performed by the developers of the qualitative agent-based models, who have a vast experience in ABMS for air traffic applications and related human factors. The mental simulation is presented for the situation where the weather at the airport suddenly and unexpectedly changes from good to bad, in both the conventional and ASAS cases. It is known from the workshop with controllers and pilots that the strategy used in such sudden weather deterioration is to reduce the runway capacity. To achieve the associated increase in spacing between the aircraft to deviate from the standard approach route according to aircraft specific directions (vectors) chosen by the controller. In order to apply the vectoring strategy, the controllers first need to cancel the airborne spacing (ASPA) operation in the ASAS case.

The analysis of interactions is presented in Section 5.1, the analysis of dynamic relations is presented in Section 5.2, and the use of the mental simulation results in the RE cycle is presented in Section 5.3.

5.1. Analysis of interactions

The initial condition of the mental simulation considers a stream of aircraft that are on approach to the runway with aircraft spacing that is commensurate with good weather. In the conventional case the aircraft spacing is maintained by the controllers using speed and vectoring instructions, whereas in the ASAS case the ASPA operation is applied to maintain the aircraft spacing. These initial conditions are extended with a sudden and considerable change in the weather at the airport, which is the varying condition studied. Table 4 shows the nominal interactions between the agents following these initial conditions for both cases (see also Fig. 4). In particular, it shows the actions of the agents and the triggers of these actions, which may be either internal or external to the agent considered. The interactions listed are the nominal actions following the initial varying condition, meaning that they do not consider other varying conditions (e.g. system failures) that may change the interactions.

In short, the following interactions occur in response to a sudden weather deterioration at the airport. If the weather suddenly deteriorates (1), the tower supervisor can decide to change the runway capacity (2). The tower supervisor coordinates with the S1/S2/ARR supervisor (3) and informs the tower controller (4). Upon this, the S1/S2/ARR supervisor may decide to adapt the airspace capacity and to cancel the ASPA operation (5), and to inform the S1/S2/ARR controllers about this decision (6).The tower controller updates the situation awareness about the runway capacity and separation standard (7). The ARR controller (8), the S1 controller (9), and the S2 controller (10) update their situation awareness about the ASPA cancellation and the new separation standard. Subsequently in the ASAS case only, these controllers use their ATC communication system to communicate (11, 12, 13, 14) the ASPA cancellation to multiple aircraft. These messages are received by the aircraft CNS systems (15) and by the pilots (16). Next, they adapt the mode of the ASAS system (17, 18) and control their aircraft speed along an approach in non-ASPA mode (19), upon which the aircraft flies the updated flight profile (20). It thus follows from this mental simulation that there is considerable increase in the number of needed agents' interactions for the ASAS case in comparison with the conventional case.

In follow-up to the initial response to a weather deterioration, wherein the runway and airspace capacities are reduced and the ASPA operation is cancelled, the controllers have to assure that the aircraft separations get in line with the new capacity requirements. It implies that they typically have to increase the spacing between the aircraft in the sequence. Interactions 21-34 in Table 4 show the agents' interactions for increasing spacing of traffic in the ARR sector; similar interactions exist for the S1 and S2 sectors. These interactions refer to aircraft 3D position and speed estimation by the ATC surveillance system, the hereupon based monitoring of aircraft states by the controller, the specification of heading instructions to increase the spacing, the transfer of the instructions, the interpretation of the instruction by the pilots and their control of the aircraft. Next, the controller keeps monitoring the aircraft position and can provide a vector back to the route if the spacing has been increased sufficiently. These kinds of interactions exist for both the ASAS case and the conventional case. The effects of these interactions strongly depend on the states of the agents. Qualitative reasoning of the dynamic evolution of such agent states is presented next in Section 5.2.

5.2. Analysis of dynamics

In addition to the analysis of interactions presented in above section, mental simulation on the basis of the qualitative agentbased model can provide qualitative insight into the dynamic relations between (aggregated) states of the approach operations in the ASAS and conventional cases. The initial condition for this mental simulation is equal to that of the analysis of interactions, being a stream of aircraft approaching in good weather either in conventional or in ASAS-supported operations. It was argued in the RE workshop that in this nominal condition the number of aircraft per timeframe that can be handled (capacity) in the ASAS case exceeds the capacity in the conventional case. This implies that the mean separation distance in the approach stream is initially smaller in the ASAS case.

The dynamics of the following aggregated states are considered in the mental simulations following this initial condition:

- The capacity (number of aircraft that can be handled in a particular timeframe), shown in Fig. 11;
- The mean separation distance of aircraft in the approach sequence (at the end of the ARR sector), shown in Fig. 11;
- The mean vectoring distance (the mean of the additional distance flown in deviation from the standard approach route as result of vectoring instructions), shown in Fig. 12;
- The mean communication load of the ARR controller (the mean fraction of time that the ARR controller uses for R/T communication with pilots), shown in Fig. 12;
- The mean task load of the ARR controller, shown in Fig. 13;
- The probability of the contextual control mode of the ARR controller (tactical or opportunistic modes), shown in Fig. 13;
- The mean separation distance at the runway threshold, shown in Fig. 14; and
- The frequency of go arounds (per approach), shown in Fig. 14.

The argumentation in support of these plots is provided next. The capacity reduction and ASPA cancellation (in the ASAS case) due to the weather deterioration at the airport, shown in Fig. 11, is the onset of all other changes in the aggregated states, shown in Figs. 11–14. The capacity is reduced more in the ASAS case than in the conventional case, since the ASAS case supports a larger

Table 4

Nominal agents' interactions in conventional and/or ASAS cases for weather deterioration leading to a reduction in runway capacity and ending of ASPA. For each interaction the trigger and resulting action are listed.

Nr	Conv	ASAS	Agent	Trigger	Action
1	×	×	Weather	Random event	Sudden and considerable weather deterioration at airport
2	×	×	TWR Supervisor	Sudden and considerable weather deterioration at	Update SA: observe and interpret weather deterioration and decide to
				airport (1)	reduce runway capacity
3	×	×	TWR Supervisor	Decision to reduce runway capacity (2)	Coordinate with S1/S2/ARR Supervisor
4	×	×	TWR Supervisor	Decision to reduce runway capacity (2)	Inform TWR Controller
5	×	×	S1/S2/ARR	Coordination with TWR Supervisor (3)	Update SA: decide on ending of ASPA operation (ASAS case only) and on
			Supervisor		airspace capacity and separation
6	×	×	S1/S2/ARR	Decision on ending of ASPA operation and	Inform S1/S2/ARR controllers about ASPA ending (ASAS case only) and
			Supervisor	separation (5)	separation standard
7	×	×	TWR Controller	Information of TWR Supervisor (4)	Update SA: runway capacity and separation standard
8	×	×	ARR Controller	Information of S1/S2/ARR Supervisor (6)	Update SA: ASPA ending (ASAS case only) and separation standard
9	×	×	S1 Controller	Information of S1/S2/ARR Supervisor (6)	Update SA: ASPA ending (ASAS case only) and separation standard
10	×	×	S2 Controller	Information of S1/S2/ARR Supervisor (6)	Update SA: ASPA ending (ASAS case only) and separation standard
11		×	ARR Controller	Update SA about ASPA and/or separation	Communicate end of ASPA operation to multiple aircraft
				standard (8)	
12		×	S1 Controller	Update SA about ASPA and/or separation	Communicate end of ASPA operation to multiple aircraft
				standard (9)	
13		×	S2 Controller	Update SA about ASPA and/or separation	Communicate end of ASPA operation to multiple aircraft
				standard (10)	
14		×	ATC Comm.	Communicate end of ASPA operation by ARR/S1/S2	Transfer communication messages
			System	controllers (11, 12, 13)	
15		×	CNS System	End-of-ASPA communication message (14)	Transfer communication messages
			Aircraft-i		
16		×	Pilots-i	End-of-ASPA communication message (15)	Update SA: ASPA operation is ended
17		×	Pilots-i	Update SA on end of ASPA (16)	Change mode of ASAS to ASPA off
18		×	ASAS-1	Mode change by pilots (17)	Turn ASPA off
19		×	Pilots-1	Update SA on end of ASPA (16)	Control aircraft along approach in non-ASPA mode
20	×	×	AIFCFAIT-I	Control aircraft in non-ASPA mode (19)	Fly along controlled trajectory
21	×	×	ATC Surveillance	Internal repeating trigger	Update aircraft position and speed data
22			ADD Comtrollor	Internal non-acting this and	Undets CA on simplify in easter using ATC surveillance data
22	×	×	ARR COllifolier	Internal repeating trigger	Deside and communicate heading to simplify and surveinance data
23	×	×	AKK CONTROLLED	Opdate SA on aircraft in sector (22)	
24		~	ATC Comm	Communicate heading to aircraft (22)	problem Transfer communication mossage
24	~	*	System	communicate nearing to ancian (23)	
25	~	~	CNS System	Heading message (24)	Transfer communication message
25	^	^	Aircraft_k	ficading incosage (24)	Hansier communication message
26	~	~	Pilots_i	Heading message (25)	Undate SA on heading requested by controller
20	Ŷ	$\hat{\mathbf{v}}$	Pilots_i	Undate SA on heading (26)	Implement heading change
27	×	Ŷ	Aircraft-i	Implement heading change (27)	Fly heading
20	×	×	ARR Controller	Update SA on aircraft in sector following heading	Provide heading back to route if separation problem is resolved
20	~	~	That controller	change (21, 23)	Torrae neuaning back to route it separation problem is resorred
30	×	×	ATC Comm.	Communicate heading to aircraft (29)	Transfer communication message
			System		
31	×	×	CNS System	Heading message (30)	Transfer communication message
			Aircraft-i	······································	······································
32	×	×	Pilots-i	Heading message (31)	Update SA on heading requested by controller
33	×	×	Pilots-i	Update SA on heading (32)	Implement heading back to route
34	×	×	Aircraft-i	Implement heading change (33)	Fly heading back to route

capacity in undisturbed circumstances. As a result of the controller strategies, the mean separation between the aircraft is expected to gradually increase following the reduction of the capacity (Fig. 11). The largest increase in the mean separation distance is achieved in the ASAS case, following the largest decrease in capacity.

Before the capacity reduction, the mean communication load of the ARR controller is expected to be less in the ASAS case than in the conventional case, since then the separation control is regulated by the ASPA system. Following the capacity reduction, a strong increase in the mean communication load (Fig. 12) is expected in the ASAS case, since firstly the end of the ASPA operation needs to be communicated to the aircraft and secondly there needs to be a considerable number of vectoring operations to increase the separation distances. In the conventional case, the increase in communication is expected to be more modest, since no ASPA operation needs to be ended and the mean separation needs to be increased to a smaller extent. When the separations in the sequence have been stabilized in correspondence with new runway capacity, the communication load is decreased to a new value that is commensurate with the vectoring operations. The mean vectoring distance in Fig. 12 shows a considerable increase in the ASAS case, since a large change in separation distance needs to be achieved, whereas a more modest bump is expected in the conventional case.

Fig. 13 shows the mental simulation results for the task load and the control mode of the ARR controller. In line with the large increase in R/T communication as explained for Fig. 12, the task load of the ARR controller is expected to increase considerably in the ASAS case, and to a larger extent than in the conventional case. In relation with this change in task load, the contextual control mode of the ARR controller is expected to mostly switch from tactical to opportunistic as a result of the capacity reduction, and this switch to the opportunistic control mode is more dominant in the ASAS case. Working in the opportunistic control mode implies in the model, that the ARR controller is performing tasks (monitoring, decision making, communication) more quickly. Although thus more tasks are performed in the opportunistic mode, the stochastic variability in the task performance has been modelled to increase in the opportunistic mode, leading to a larger likelihood of tasks not being performed as they should or as intended.



Fig. 11. Mental simulation results for the ASAS and conventional cases, for a situation of weather deterioration at the airport. Top row shows the decrease in capacity. Bottom row shows the mean separation distance of aircraft in the approach sequence.



Fig. 12. Mental simulation results for the ASAS (left) and conventional (right) cases, for a situation of weather deterioration at the airport. Top row shows the mean vectoring distance. Bottom row shows the mean communication load of the ARR controller.

Fig. 14 shows the mental simulation results for the mean separation at the runway threshold and the frequency of go arounds instructed by the TWR controller. It is expected that the frequency of go arounds increases, since the required increase in separation is not always achieved. Moreover, the frequency of go arounds is expected to be higher in the ASAS case, since in this case the capacity reduction is larger and the S1, S2 and ARR controllers have a dual job of both ending the ASPA operation and increasing the separation.

5.3. Use of mental simulation results in the RE cycle

The mental simulation results for the interactions and dynamic relations presented in above sections were derived on the basis of the qualitative agent-based model and supported by knowledge of the strategies used for the deteriorated weather condition achieved in a workshop with controllers and pilots. The types of interactions in the agent-based model, and the mental simulation results for the increase in vectoring, task load and go arounds following the runway capacity decline are in line with the strategies and expectations expressed during the workshop. However, the interactions and details of the dynamic relations were not discussed with pilots and controllers during a follow-up workshop, nor were other types of validation exercises done for these results.

Given the uncertainty in the current results of the mental simulations, they can best be formulated as hypotheses, as a basis for further research. The following hypotheses regarding the adaptive capacity of the sociotechnical system of the approach operations in dealing with the sudden weather deterioration have been identified.



Fig. 13. Mental simulation results for the ASAS (left) and conventional (right) cases for the situation of strong weather deterioration at the airport. Top row shows the mean task load of the ARR controller. Bottom row shows probability of attaining tactical/opportunistic control modes by the ARR controller.



Fig. 14. Mental simulation results for the ASAS (left) and conventional (right) cases for the situation of strong weather deterioration at the airport. Top row shows the mean separation distance at the runway threshold. Bottom row shows the frequency of go arounds as instructed by the TWR controller.

- The varying condition is expected to imply a considerable increase in the task load of the ARR, S1 and S2 controllers in the ASAS case, since they need to communicate the end of the ASPA operation to the aircraft in their sectors, as well as to decide on and communicate vectors to their aircraft so as to assure the increased separation standard.
- Such increased task load has impact on the way of working of the ARR, S1 and S2 controller. In reference to the contextual control mode model of Hollnagel (1993) this may lead to a larger probability of working in an opportunistic control mode and to more deviations in task performance. Such deviations have to be recognized and dealt with somehow by the sociotechnical system, thus stretching the adaptive capacity.
- It is expected that the varying condition leads to an increase in the frequency of go arounds, especially in the ASAS case. The aircraft being sent around have to be added to the approach sequence once again, thus putting further demand on the adaptive capacity.

As part of the RE cycle of Fig. 1, above hypotheses may be studied by using the approaches introduced in Section 2.3.3. A key approach for a more profound analysis is the further development, parameter quantification, and computer simulation of a quantitative agent-based model. This will allow a deeper understanding of the relations and the timing implications on the performance of the sociotechnical system. It will allow getting quantitative results for the graphs such as in Figs. 11–14, and more. Such quantitative agent-based modelling and simulation may also be used to determine an appropriate maximum capacity of ASPA operations, such that the sociotechnical system can effectively attain lower capacities in the case of sudden weather changes. The attained capacity value may be validated by human-in-the-loop simulations for the varying condition considered.

If the identified hypotheses would be verified in additional research, they could be used to identify recommendations to improve the design of the ASAS operation for dealing resiliently with the bad weather situation. If so, such design recommendations could include:

- Find ways to reduce the task load of ARR, S1 and S2 controllers in the situation that the ASPA operation needs to be cancelled and the capacity needs to be reduced. For instance, by adding a temporary assistant, or by developing a tool for quickly communicating end of ASPA operation to aircraft.
- Use a maximum capacity in ASPA operations which can be effectively downsized in the case of sudden runway capacity reduction.
- Assure that aircraft going around can be effectively integrated in the approach stream.

6. Discussion

Following a discussion of some recent reviews we identified the need in RE for more structured modelling approaches for analysis of resilience in sociotechnical systems that can support both qualitative and quantitative studies. In this paper we presented ABMS as an approach towards this need. Although in general ABMS is considered to be a quantitative approach, in this paper we showed that development of an agent-based model starts with a qualitative development step and that mental simulation can be used to obtain qualitative results. Such qualitative analysis was illustrated in detail for aircraft approach operations using conventional systems and ASAS.

An agent-based model of a sociotechnical system represents a collection of human agents and technical system agents which interact among each other and with the environment in which they reside. The agents have time-dependent states, inputs and outputs, and the evolution of these states, the impact of the input signals on the states, and the implications of the states for the output signals are represented by sets of model constructs. In the application case, eleven model constructs were used, which were chosen from a library of model constructs that was developed for ATM sociotechnical systems. For the human agents these model constructs included multi-agent situation awareness, as a key construct for the agent's situation awareness in a multi-agent environment, several task-related (identification, scheduling, execution, decision making) model constructs, task load and contextual control mode as workload-related model constructs, and variability-related model constructs representing dynamics, stochasticity and errors in human performance. For the technical system agents smaller sets of model constructs were used, including multi-agent situation awareness, decision making, system mode, and dynamic and stochastic variability. Overall, the detailed qualitative description along these agent-based model constructs provides a well-defined and structured overview of the information flows, task work done, workload, and dynamic and stochastic performance variability of the humans, as well as of the information flows and performance of the technical systems in the operation studied. As such, the achieved agent-based model is an analysis result in itself, which structures the understanding of the task performance, interactions, and the inherent variability of the operations.

The developed agent-based model was used for structured qualitative reasoning about the dynamic interactions between the agents and their effects on agents' variables. This kind of reasoning is termed 'mental simulation' as it is completely based on specialist model-based reasoning, without using computer implementation and simulation of the agent-based model. Using such mental simulation for the modelled conventional and ASAS operations, we identified several differences between the operations for dealing with varying conditions. These differences refer to the number and kinds of interactions between the human and technical system agents, and to the dynamic evolution of performance variables, such as the mean separation, the vectoring distance, the task load, the contextual control mode, and the number of go arounds during final approach. As thus illustrated, mental simulation can provide insight into the sociotechnical system performance and provide feedback for design and additional analysis.

The development of a qualitative agent-based model and the mental simulation using this model are part of an RE approach for analysis and improvement of the resilience of a sociotechnical system. Main methods in this RE cycle are workshops with human operators who are proficient in the operations studied. Workshops are common in RE approaches and they can effectively provide a broad range of strategies used by operators in all kinds of contextual conditions. The analysis towards a proper understanding of how the combined strategies and behaviours of interacting human operators and technical systems have impact on the adaptive capacity of a sociotechnical system for dealing with ranges of varying conditions is however far more difficult. As such, for more complex operations, analysis results tend to be high level only, or very uncertain with respect to performance indicators, or results cannot be obtained at all by workshop centred RE approaches. The qualitative ABMS approach is a first step towards a more structured analysis, which is supported both by the development of a qualitative agent-based model and by mental simulation. It was illustrated in detail in this paper how insights can thus be obtained for air traffic operations in varying conditions.

In the application case we focused on a particular sudden bad weather condition that influences aircraft approach operations, we used qualitative agent-based modelling to arrive at the hypothesis that operations in the ASAS case for this condition are less resilient than in the conventional case, and next we presented some options to improve the resilience of the operation in the ASAS case for the sudden bad weather condition. These steps were all part of an RE cycle that provides feedback to design in order to improve resilience. Although we argued that the resilience for dealing with the sudden bad weather condition is less in the ASAS case, this does not imply that overall the resilience of the ASAS operations would necessarily be less. We did not study the overall level of resilience and we recognize that considerably more research would be needed to arrive at conclusions on this. Questions within such research would include the following: What is the overall set of varying conditions for studying the resilience of the operation? How often do these varying conditions occur? What is the impact of the strategies for dealing with these varying conditions on performance indicators of interest? How can these various impacts on performance indicators be combined for all varying conditions to come at statements on the overall resilience of the sociotechnical system?

The understanding that the overall performance of a complex sociotechnical system can be highly variable is one of the foundations of the RE field. This overall performance variability is due to the intrinsic stochastic dynamics of the elements of a sociotechnical system and its environment, and due to all their interactions. The overall performance can be regarded as emergent, as it cannot be understood by its single elements, but only by considering the totality of all elements and interactions. The description of its elements and interactions in the qualitative agent-based model supports analysts to reason in a structured way about the overall performance of the sociotechnical system. It is clear, nevertheless, that such mental simulation is only a first step towards understanding the overall performance. A principal limitation of mental simulation is that it is less feasible for models with more complex interactions to keep track of all possible implications of stochastic variations, dynamics and interactions of the agents that may follow a particular initial condition. Mental simulation results are therefore bound to mainly describe selected aggregated variables at a high level for more usual modes of working and interactions, and to neglect details in variations and less usual performance aspects. If it can be argued in cooperation with stakeholders, that the level of the uncertainty in the mental simulation results is commensurate with the objectives of the RE cycle and no quantitative results are needed, then the mental simulation results can be accepted as a final result of Step 3 of the RE cycle and used as a basis to identify improvements in resilience. If such uncertainty is not acceptable or quantitative results are needed, then the next step towards development, implementation and computer simulation of a quantitative agent-based model should be pursued. Computer simulations can overcome the limitations of mental simulation by evaluating in detail all stochastic variations, dynamics and interactions in the agent-based model. Another advantage of computer simulations is that they can be completely tractable. in contrast with the reasoning of an analyst during mental simulations or the reasoning of operational experts during a workshop.

Quantitative ABMS steps have not been addressed in detail in this paper, as our objectives are to present the principles of ABMS as a way to support RE and to show how ABMS can provide qualitative results. The follow-up steps for quantitative ABMS can in general be pursued using various methods and tools, such as presented in overviews of Macal and North (2010) and Van Dam et al. (2013). A specific way how quantitative ABMS for RE may be achieved is by using methods from agent-based dynamic risk modelling (DRM) (Blom et al., 2006; Everdij et al., 2014). Agentbased DRM has used model constructs similar to those shown in this paper, model formalization by dynamically coloured Petri nets, and Monte Carlo simulations for estimating accident probabilities of air traffic scenarios. A key difference is that agent-based DRM requires rare event estimation techniques to assess the very low accident probabilities of air traffic scenarios, whereas ABMS for RE focuses on understanding effects of strategies and adaptive capacity for the performance of the sociotechnical system in the context of varying conditions. To arrive at the low accident probabilities in agent-based DRM, specialist knowledge on rare event estimation and typically long computation times are needed. ABMS for RE does not need the many replications in Monte Carlo simulations, since it is not focused on estimation of very rare events. As such, the scenarios and aspects of the sociotechnical system considered in ABMS for RE may be more extensive and represented in more detail. In the light of the emphasis on everyday actions and outcomes in Safety-II (Hollnagel, 2014), this means that these may be modelled and understood in more detail using ABMS for RE. Thus achieved insight into the most important aspects of the sociotechnical may next be used in more dedicated models as part of agent-based DRM for assessment of accident probabilities. Such connection between ABMS for RE and agent-based DRM needs to be studied in follow-up research.

A key question for using results of ABMS in the RE cycle is: what is their validity? In other words, how well do the models and simulation results represent reality, within the specific context for which the model was developed? We like to discuss validity using the concept of uncertainty, i.e. the extent by which something is not certain. We already discussed one source of uncertainty, which is especially relevant for the qualitative ABMS phase, being the uncertainty due to limitations of the mental simulation process. Another main source of uncertainty is lack of complete knowledge of elements and their interactions in the sociotechnical system and/or limitation of the ways that they are represented in the agent-based model. In the qualitative ABMS phase this concerns the high-level specification via model constructs and their interconnections. In the quantitative ABMS phase this would concern the structure of the formal models and the quantification of their parameters. It is clear that the level of uncertainty in models of sociotechnical systems can be large, as many of their aspects are often not known precisely. In the RE domain, conceptual models are frequently used, which by their nature are generic and therefore uncertain for a particular sociotechnical system. A qualitative agent-based model, such as illustrated in this paper, provides more detail for the interacting agents and their constituent models and thereby may describe the performance of the sociotechnical system with more certainty. Nevertheless, it is clear that a qualitative model primarily describes types of mechanisms and interactions, and that the level of uncertainty in the achieved mental simulation results may well be significant. The most accurate results may be achieved by a quantitative agent-based model in combination with computer simulation. Whether results with low uncertainty can actually be realized in the quantitative ABMS phase depends on the validity of the individual agents' models, the validity of the integrated set of all models for the operational context, and the appropriateness of the parameter values. There are various ways to evaluate the level of uncertainty in these components and in the overall results, including comparison with results in the literature, comparison with available measurement data of related experiments or real operations, and discussion with operational experts in the RE cycle. Quantitative ABMS can also effectively support analysis of the sensitivity of performance indicators for parameter variations, as a way to gain an understanding of the relative importance of model parts for performance indicators of interest. The ways that quantitative ABMS can most effectively be applied in RE needs to be studied in detail in follow-up research.

How do ABMS and mental simulation compare with FRAM (Hollnagel, 2012)? Some high-level observations are provided next, but we recognize that a more detailed comparison would be needed to arrive at more definitive conclusions. The functional analysis-based approach of FRAM is structured by a single model construct, which describes input, output, time, control, preconditions and resources for each function or activity in a sociotechnical system. An advantage of this single model construct is that it can be easily explained to a wide audience, but main limitations are (1) it is very abstract, such that it may be difficult to represent nuances of a sociotechnical system, and (2) the function-based focus does not explicitly account for the boundaries between entities (humans, technical systems) in the sociotechnical systems, nor for their dynamic states. This can be contrasted with the library of model constructs in ABMS, which enables more nuanced and detailed modelling, and its agent perspective, which explicitly accounts for the boundaries between entities and which is statebased. FRAM employs mental simulation as its principal evaluation means. This model-based reasoning is primarily focused on arguing about interactions between functions rather than on the dynamics of sociotechnical systems.

A fundamental question is to what extent detailed and possibly quantitative models can be used to study and improve resilience. Woods (2015) stresses that key for resilience is the ability of a system to deal with surprise events. Woods uses concepts as graceful extensibility and sustained adaptability to discuss the capacity of a sociotechnical system to deal effectively with surprises. Two types of surprise events can be distinguished: (1) surprises for the agents but known to the modeller, and (2) surprises unknown to the modeller. Type 1 surprises can be included in an agent-based model, where agents possess particular sets of strategies for a range of varying conditions that are known by the agent, but which exclude the type 1 surprises. Next it can be analysed what the effect is of the agent-based model for the surprise event, such as the rebound of robustness with respect to the event. In this way insight is gained into the capacity of the modelled sociotechnical system to deal with a particular type 1 surprise and this knowledge can be used to improve the resilience of the actual system. Type 2 surprises pose a difficulty for modelling, as these are not a priori

known and thus cannot be modelled. What can be done for type 2 surprises, however, is to evaluate the capability of the agent-based model for dealing with large sets of different type 1 surprises in combination with other varying conditions, and to use this knowledge to improve the sociotechnical system. As long as there is some similarity of type 2 surprises with the studied set, this may help to attain a more resilient system also for these surprises. It can be recognized that in these analyses for type 1 and type 2 surprises, system responses such as considered in the rebound and robustness concepts are used as bridges towards attaining resilience in the graceful extensibility and sustained adaptability concepts of Woods (2015).

In conclusion, ABMS is a detailed analysis approach for studying resilience of sociotechnical systems that offers a flexible range of model constructs enabling representation of the work-as-done in the system. The agent-based perspective fits well with usual views on elements of a sociotechnical system and it naturally couples states and behaviour of the agents. Simulation in ABMS can be done mentally, leading to qualitative results as shown in this paper, or it can be pursued by computer simulation. The capability to evaluate many interacting varying conditions by computer simulations sets open the door to truly studying performance variability in normal situations, as envisioned in Safety-II, rather than focusing on some restricted non-nominal cases, as typically addressed in Safety-I. This will be further pursued in follow-up research.

Acknowledgements

The authors thank the anonymous reviewers for their very useful comments and insights, which helped to considerably improve the paper. The authors would like to thank the organisations and participants who supported the use cases and test applications. The research reported upon was funded as part of the SESAR Joint Undertaking within P16.06.01b. The views and opinions in this publication are of the authors and are not intended to represent the positions of SESAR JU or its project member organisations.

References

- Baumeister, R.F., Masicampo, E.J., 2010. Conscious thought is for facilitating social and cultural interactions: how mental simulations serve the animal-culture interface. Psychol. Rev. 117 (3), 945–971. http://dx.doi.org/10.1037/a0019393.
- Bergström, J., van Winsen, R., Henriqson, E., 2015. On the rationale of resilience in the domain of safety: a literature review. Reliab. Eng. Syst. Saf. 141, 131–141. http://dx.doi.org/10.1016/j.ress.2015.03.008.
- Blom, H.A.P., Daams, J., Nijhuis, H.B., 2001. Human cognition modelling in ATM safety assessment. In: Donohue, G.L., Zellweger, A.G. (Eds.), Air Transport Systems Engineering, AIAA, Washington D.C., USA, pp. 481–511.
- Blom, H.A.P., Stroeve, S.H., De Jong, H.H., 2006. Safety risk assessment by Monte Carlo simulation of complex safety critical operations. In: Redmill, F., Anderson, T. (Eds.), Developments in Risk-Based Approaches to Safety: Proceedings of the Fourteenth Safety-Critical Systems Symposium, Bristol, U.K., 7–9 February 2006. Springer.
- Blom, H.A.P., Bakker, G.J., 2012. Can airborne self separation safely accommodate very high en-route traffic demand? In: Proceedings AIAA ATIO Conference, Indianapolis, Indiana, USA.
- Bonabeau, E., 2002. Agent-based modeling: methods and techniques for simulating human systems. Proc. Natl. Acad. Sci. USA 99 (3), 7280–7287.
 Chandrasekharan, S., Nersessian, N.J., Subramanian, V., 2013. Computational
- Chandrasekharan, S., Nersessian, N.J., Subramanian, V., 2013. Computational modeling: is this the end of thought experiments in science? In: Brown, J., Frappier, M., Meynell, L. (Eds.), Thought Experiments in Philosophy, Science and the Arts. Routledge, London.Chen, B., Cheng, H.H., 2010. A review of the applications of agent technology in
- Chen, B., Cheng, H.H., 2010. A review of the applications of agent technology in traffic and transportation systems. IEEE Trans. Intell. Transp. Syst. 11 (2), 485– 497.
- Clement, J., 2008. Creative Model Construction in Scientists and Students: The Role of Imagery, Analogy, and Mental Simulation. Springer Science & Business Media.

- Dekker, S., 2005. Ten Questions About Human Error: A New View of Human Factors and System Safety. Lawrence Erlbaum Associates, London, UK.
- Endsley, M.R., 1995. Toward a Theory of Situation Awareness in Dynamic Systems. Human Factors: J. Human Factors Ergon. Soc. 37 (1), 32–64. http://dx.doi.org/ 10.1518/001872095779049543.
- Everdij, M., Stroeve, S., Bottone, M., Diaz Dominguez, C., De Gelder, N., Paino, M., 2016. ASAS - Interim Report 2014 (Edition 2.0 ed.). Project 16.06.01b, Deliverable D03–001. SESAR.
- Everdij, M.H.C., Blom, H.A.P., Stroeve, S.H., Kirwan, B., 2014. Agent-based dynamic risk modelling for ATM: A white paper: Eurocontrol. http://www.nlr-atsi. nl/downloads/agent-based-dynamic-risk-modelling-for-atm.pdf>.
- Francis, R., 2013. Analysis of resilience in manmade and natural systems. Deliverable D1.1. Resilience2050.eu.
- Francis, R., Bekera, B., 2014. A metric and frameworks for resilience analysis of engineered and infrastructure systems. Reliab. Eng. Syst. Saf. 121, 90–103. http://dx.doi.org/10.1016/j.ress.2013.07.004.
- Hegarty, M., 2004. Mechanical reasoning by mental simulation. Trends Cogn. Sci. 8 (6), 280–285.
- Herrera, I.A., Woltjer, R., 2010. Comparing a multi-linear (STEP) and systemic (FRAM) method for accident analysis. Reliab. Eng. Syst. Saf. 95 (12), 1269–1275. http://dx.doi.org/10.1016/j.ress.2010.06.003.
- Holling, C.S., 1973. Resilience and stability of ecological systems. Ann. Rev. Ecol. Syst. 4, 1–23.
- Hollnagel, E., 1993. Human Reliability Analysis, Context and Control. Academic Press, London, UK.
- Hollnagel, E., Woods, D.D., Leveson, N., 2006. Resilience Engineering: Concepts and Precepts. Ashgate, Aldershot, England.
- Hollnagel, E., 2012. FRAM: The Functional Resonance Analysis Method; Modelling Complex Socio-Technical Systems. Ashgate, Farnham, England.
- Hollnagel, E., 2014. Safety-I and Safety-II: The Past and Future of Safety Management. Ashgate, Farnham, UK.
- Hosseini, S., Barker, K., Ramirez-Marquez, J.E., 2016. A review of definitions and measures of system resilience. Reliab. Eng. Syst. Saf. 145, 47–61. http://dx.doi. org/10.1016/j.ress.2015.08.006.
- Macal, C.M., North, M.J., 2010. Tutorial on agent-based modelling and simulation. J. Simul. 4, 151–162.
- Macchi, L., Hollnagel, E., Leonhardt, J., 2011. Resilience engineering approach to safety assessment: An application of FRAM for the MSAW system. hal-00572933, version 1. Mines ParisTech.
- Markman, K.D., Klein, W.M., Suhr, J.A., 2012. Handbook of Imagination and Mental Simulation. Psychology Press.
- Martin-Breen, P., Anderies, J.M., 2011. Resilience: A Literature Review. The Bellagio Initiative.
- Nersessian, N.J., 1992. In the theoretician's laboratory: Thought experimenting as mental modeling. In: PSA: Proceedings of the Biennial Meeting of the Philosophy of Science Association, pp. 291–301.
- Nersessian, N.J., 1999. Model-Based Reasoning in Conceptual Change. Model-Based Reasoning in Scientific Discovery. Springer.Paulo Victor Rodrigues, d.C., 2011. The use of Functional Resonance Analysis
- Paulo Victor Rodrigues, d.C., 2011. The use of Functional Resonance Analysis Method (FRAM) in a mid-air collision to understand some characteristics of the air traffic management system resilience. Reliab. Eng. Syst. Saf. 96 (11), 1482– 1498. http://dx.doi.org/10.1016/j.ress.2011.05.009.
- Pinska-Chauvin, E., Josefsson, B., Branlat, M., Everdij, M., Stroeve, S., Laursen, T., Herrera, I.A., 2016. T4 Final report. Project 16.06.01b, D04. SESAR Joint Undertaking.
- Praetorius, G., Hollnagel, E., Dahlman, J., 2015. Modelling Vessel Traffic Service to understand resilience in everyday operations. Reliab. Eng. Syst. Saf. 141, 10–21. http://dx.doi.org/10.1016/j.ress.2015.03.020.
- Stroeve, S.H., Blom, H.A.P., Van der Park, M.N.J., 2003. Multi-agent situation awareness error evolution in accident risk modelling. In: Proceedings 5th USA/Europe Air Traffic Management R&D Seminar, Budapest, Hungary.
- Stroeve, S.H., Blom, H.A.P., Bakker, G.J., 2013a. Contrasting safety assessments of a runway incursion scenario: event sequence analysis versus multi-agent dynamic risk modelling. Reliab. Eng. Syst. Saf. 109, 133–149. http://dx.doi. org/10.1016/j.ress.2012.07.002.
- Stroeve, S.H., Bosse, T., Blom, H.A.P., Sharpanskykh, A., Everdij, M.H.C., 2013b. Agent-Based Modelling for Analysis of Resilience in ATM. Third SESAR Innovation Days, Stockholm, Sweden.
- Sun, R. (Ed.), 2006. Cognition and Multi-Agent Interaction: From Cognitive Modeling to Social Simulation. Cambridge University Press, New York, USA.
- Van Dam, K.H., Nikolic, I., Lukszo, Z., 2013. Agent-Based Modelling of Socio-Technical Systems. Springer, Dordrecht, The Netherlands.
- Waytz, A., Hershfield, H.E., Tamir, D.I., 2015. Mental simulation and meaning in life. J. Pers. Soc. Psychol. 108 (2), 336–355.
- Woods, D.D., 2015. Four concepts for resilience and the implications for the future of resilience engineering. Reliab. Eng. Syst. Saf. 141, 5–9. http://dx.doi.org/ 10.1016/j.ress.2015.03.018.
- Wooldridge, M., Jennings, N.R., 1995. Intelligent agents: theory and practice. Knowl. Eng. Rev. 10 (2), 115–152.